Author's response to reviews

Title: A fatal pseudo-tumour: disseminated basidiobolomycosis

Authors:

Guido GEL Van den berk (g.e.vandenberk@hotmail.com)
Arnold L.A. Noorduyn (L.A.Noorduyn@amc.uva.nl)
Ruud R.J. van Ketel (R.J.vanketel@amc.uva.nl)
Jeannouel J van Leeuwen (guidovdberk@hotmail.com)
Willem W.A. Bemelman (W.A.Bemelman@amc.uva.nl)
Jan J.M. Prins (J.M.Prins@amc.uva.nl)

Version: 5 Date: 25 August 2006

Author's response to reviews:

At the end of the abstract,

contributions of all authors are described in detail, with the statement that none of the authors have any competing interests, financial or non-financial.

Furthermore,

an acknowledgement is made to the wife of the patient described, who gave written informed consent for publication of this case report.